Author’s response to reviews

Title: Dental caries prevention strategies among children and adolescents with immigrant- or low socioeconomic backgrounds- do they work? A systematic review

Authors:

Marit Slåttelid Skeie (marit.skeie@odont.uib.no)

Kristin S Klock (kristin.klock@uib.no)

Version: 3 Date: 14 Dec 2017

Author’s response to reviews:

Bergen, Monday, December 14, 2017

Editorial Board, BMC Oral Health

We hereby submit the new revised manuscript of "Dental caries prevention strategies among children and adolescents with immigrant- or low socioeconomic backgrounds- do they work? A systematic review" for the final publication in BMC Oral Health. The points to be addressed, have been done (see below).

Point 1: The statement “communicable” is removed (the first sentence in the document).

Point 2: Under the “Ethics Approval, it now stands “Not Applicable”.

Point 3: Both authors’ addresses are on the title page.

Point 4: All track changes are removed, and in this version of manuscript, only black text is used.

Yours sincerely,

Marit Slåttelid Skeie

(on behalf of the authors).